Author’s response to reviews

Title: Determinants of intentions to monitor antihypertensive medication adherence in Irish community pharmacy: a factorial survey

Authors:

Paul Dillon (pauldillon@rcsi.ie)
Ronald McDowell (ronaldmcdowell@rcsi.ie)
Susan Smith (susansmith@rcsi.ie)
Paul Gallagher (pgallagher@rcsi.ie)
Gráinne Cousins (gcousins@rcsi.ie)

Version: 3 Date: 26 Aug 2019

Author’s response to reviews:

Dear Editor and Reviewers,

Thank you for your time in extensively reviewing my manuscript. Please see below my response to editorial comments.

Best wishes,

Editor Comments:

1. Please provide a List of abbreviations after the Conclusions section. If abbreviations are used in the text, they should be defined in the text at first use and included in this list.

Abbreviations included
2. In the Ethics approval and consent to participate section of the Declarations, please state the form of consent obtained from participants.

Electronically declared informed consent added

3. As Susan Smith is a member of the editorial board (Associate Editor) of this journal, in order to ensure transparency, please declare this in the Competing Interests section of the Declarations.

Prof Smith's role as editorial board member has been added to competing interests section

4. We noticed that two of the supplementary files in the file inventory are titled Appendix 2. Please change one of the Descriptions to Appendix 3 and ensure that they are correctly referenced in the main text of the manuscript.

Corrected

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colors. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Done